Author's response to reviews

Title: Combinations of Motor Measures More Strongly Predict Adverse Health Outcomes in Old Age: The Rush Memory and Aging Project, a Community-Based Cohort Study.

Authors:

Aron S Buchman (aron_s_buchman@rush.edu)
Sue E Leurgans (Sue_E_Leurgans@rush.edu)
Patricia A Boyle (Patricia_Boyle@rush.edu)
Julie A Schneider (Julie_A_Schneider@rush.edu)
Steven A Arnold (steven.arnold@uphs.upenn.edu)
David A Bennett (dbennett@rush.edu)

Version: 2 Date: 8 February 2011

Author's response to reviews:

There were no comments to respond to.